Author's response to reviews

Title: Feasibility of Real-Time Three-Dimensional stress echocardiography: pharmacological and semi-supine exercise

Authors:

Lorenza Pratali (lorenza@ifc.cnr.it)
Sabrina Molinaro (sabrina.molinaro@ifc.cnr.it)
Anca I Corciu (anca.corciu@ifc.cnr.it)
Emilio Pasanisi (pasanisi@ifc.cnr.it)
Marco Scalese (marco.scalese@ifc.cnr.it)
Rosa Sicari (rosas@ifc.cnr.it)

Version: 2 Date: 9 March 2010

Author's response to reviews:

Here is enclosed the revised manuscript as request by reviewers.
Page 3: line 3 first is firstly;
line 4 second is secondly;
line 13 supine is semisupine
Page 3, Patient population “Indication for stress echocardiography was suspected coronary artery disease and risk stratification of known coronary artery disease” is changed as: Indication for stress echocardiography was suspected coronary artery disease.
Page 5: line 3 end diastolic volume is end-diastolic volume and end systolic volume is end-diastolic volume. We added 2 movies as figure 1 and 2.
All changes is in the revised manuscript in yellow.